Author's response to reviews

Title: Perceptions about hemodialysis and transplantation among African American adults with end-stage renal disease: inferences from focus groups

Authors:

Megan L Salter (meg@jhmi.edu)
Komal Kumar (komalkumar91@gmail.com)
Andrew Law (alaw6@jhu.edu)
Natasha Gupta (ngupta21@jhmi.edu)
Kathryn Marks (ktmac207@gmail.com)
Kamna Balhara (kamna.balhara@gmail.com)
Mara McAdams-DeMarco (mmcadam4@jhmi.edu)
Laura Taylor (Laura.Taylor@jhu.edu)
Dorry L Segev (dorry@jhmi.edu)

Version: 4 Date: 25 March 2015

Author's response to reviews:

Dear Ms. Maria Merrie Jul Ladag and Dr Kerri Cavanaugh,

We have made the requested changes.

Sincerely,

Megan Salter, PhD, on behalf of all co-authors